Author's response to reviews

Title: Histological type and grade of breast cancer tumors by parity, age at birth, and time since birth: a register-based study in Norway

Authors:

Grethe Albrektsen (msega@uib.no)
Ivar Heuch (ivar.heuch@math.uib.no)
Steinar Ø Thoresen (steinar.o.thoresen@gsk.com)

Version: 4 Date: 12 May 2010

Author's response to reviews:

Correcting typewriting errors in page proof -
ref. no. 28, 'pregnancies' replaced by 'childbirths'
ref. no. 33, pages corrected (from 616-623 to 620-621)
ref. no. 43, bold for No. in journal
ref. no. 44, 'cancer' replaced by 'cancers'